Author's response to reviews

Title: The efficacy of physiotherapy upon shoulder function following axillary dissection in breast cancer, a pilot study.

Authors:

Carien H.G. Beurskens (c.beurskens@fysio.css.umcn.nl)
Caro J.T. van Uden (cvanuden@surgery.azm.nl)
Luc J.A. Strobbe (ljiastrobbe@cwz.nl)
Rob A.B. Oostendorp (roostendorp@kwazo.umcn.nl)
Theo Wobbes (t.wobbes@chir.umcn.nl)

Version: 2 Date: 7 March 2007

Author’s response to reviews:

Dear Sir, Madam,

We have uptained a trail registration number. This number is ISRCTN31186536.

Kind regards,

Caro van Uden